Author's response to reviews

Title: Hardiness Personality and Burnout Syndrome among nursing students of three Brazilian universities - An analytical study

Authors:

Rodrigo M Silva Rn (marques-sm@hotmail.com)
Carolina T Goulart Msc (carolinatonini@yahoo.com.br)
Luis Felipe D Lopes Prof (lflopes67@yahoo.com.br)
Patrícia M Serrano Msc (paserrano@usp.br)
Ana Lúcia S Costa Prof (ananascosta@usp.br)
Laura A Guido Prof (lguido344@gmail.com)

Version: 3 Date: 20 July 2013

Author's response to reviews:

First, we describe Acknowledgement section into manuscript, including: writing revision services (English review, as well as statistical and scientific revisions) and financial support which will be provided if this paper is accepted in the future. We highlight that all phases of research was financially supported by authors, including translations and data analysis.

Concerning to English review, a native-speaker and a sworn translator performed this job.

Also, titles and additional files were removed from the system.